**Author's response to reviews**

**Title:** The effect on survival of continuing chemotherapy to near death

**Authors:**

- Akiko M Saito (amoriya-tky@umin.ac.jp)
- Mary Beth Landrum (landrum@hcp.med.harvard.edu)
- Bridget A Neville (Bridget_Neville@dfci.harvard.edu)
- John Z Ayanian (ayanian@hcp.med.harvard.edu)
- Craig C Earle (craig.earle@ices.on.ca)

**Version:** 6  **Date:** 13 September 2011

**Author's response to reviews:**

Slight edit to one sentence in the abstract to clarify meaning.